Author's response to reviews

Title: The value of age and medical history for predicting colorectal cancer and adenomas in people referred for colonoscopy.

Authors:

Barbara-Ann Adelstein (b.adelstein@unsw.edu.au)
Petra Macaskill (petra.macaskill@sydney.edu.au)
Robin M Turner (robin.turner@sydney.edu.au)
Peter H Katelaris (peter.katelaris@sydney.edu.au)
Les Irwig (les.irwig@sydney.edu.au)

Version: 5 Date: 23 August 2011

Author's response to reviews:

Dear Editor,

As requested, I have moved all the tables form the text to the end of the manuscript. I have also re-uploaded figure 1 which was not oriented correctly. I hope that this means that all changes required are now made!

Yours sincerely

Barbara-Ann Adelstein